Author's response to reviews

Title: Cataract subtype risk factors identified from the Korea National Health and Nutrition Examination survey 2008 -2010

Authors:

Tyler Hyung Taek Rim (eyedoc.htrim@gmail.com)
Min-hyung Kim (vigor2mh@gmail.com)
Woon Cho Kim (wckim@emory.edu)
Tae-Im Kim (TIKIM@yuhs.ac)
Eung Kweon Kim (eungkkim@yuhs.ac)

Version: 3 Date: 26 December 2013

Author's response to reviews:

We fully revised our paper and thank the reviewers
"#" means that we changed the manuscripts as "before # after"

Reviewer's report:
In my opinion, the revised manuscript is done well and improved. My questions were answered adequately.
But, I still suggest authors to rephrase the Method regarding the slit lamp examination.
For example: Without iridodilator usage, the characteristics of lens were assessed using slit lamp with proper brightness, height, and width.
## We changed the text below.
The slit lamp was set at the maximal height and minimal width in order to induce dilated pupil without the need for iridodilators.
# Without iridodilator usage, the characteristics of lens were assessed using slit lamp with proper brightness, height, and width.

In addition, please check some type errors.
### The errors have been modified.

Thank you.